Author's response to reviews

Title: Osseointegration of zirconia implants: an SEM observation of the bone-implant interface

Authors:

Rita Depprich (depprich@med.uni-duesseldorf.de)
Holger Zipprich (depprich@med.uni-duesseldorf.de)
Michelle Ommerborn (ommerborn@med.uni-duesseldorf.de)
Eduardo Mahn (depprich@med.uni-duesseldorf.de)
Lydia Lammers (depprich@med.uni-duesseldorf.de)
Jörg Handschel (depprich@med.uni-duesseldorf.de)
Christian Naujoks (depprich@med.uni-duesseldorf.de)
Hans-Peter Wiesmann (depprich@med.uni-duesseldorf.de)
Norbert R. Kübler (depprich@med.uni-duesseldorf.de)
Ulrich Meyer (depprich@med.uni-duesseldorf.de)

Version: 2 Date: 10 October 2008

Author's response to reviews: see over
Dear Dr. Stamm,

Please find enclosed our revised manuscript # 1669557614214393 entitled “Osseointegration of zirconia implants: an SEM observation of the bone-implant interface” submitted to Head and Face Medicine.

Thank you ever so much for the opportunity to improve our manuscript. We really hope that you will be satisfied with our revisions and you will find it now suitable for publication in your journal.

Thanking you in anticipation.

Yours sincerely,

Dr. Michelle A. Ommerborn